Author's response to reviews

Title: Burden of Leprosy in Malawi: Community camp-based Cross-sectional Study

Authors:

Kelias P Msyamboza (msyambozak@mw.afro.who.int)
Leonard R Mawaya (mawayaleonard@yahoo.co.uk)
Hudson W Kubwalo (kubwaloh@mw.afro.who.int)
David Ng’oma (davidfidess@yahoo.com)
Moses Liabunya (moselia2002@yahoo.com)
Sunganani Manjolo (manjolos@yahoo.com)
Pilirani Msiska (pmsiska@yahoo.com)
Wilson W Somba (wwsomba@yahoo.com)

Version: 5 Date: 14 July 2012

Author's response to reviews: see over
Comment- by – comment response to reviewers

Reviewer’s report

Title: Burden of Leprosy in Malawi: Population-based Cross-sectional Study

Version: 4 Date: 10 July 2012

Reviewer: ANIL KUMAR

Reviewer’s report:
The revised version has improved but does not comply with population based estimate. I think authors cannot improve it further due to limitations, I therefore feel the paper may be accepted for publication if authors agree for slight change in its title "Burden of leprosy in Malawi: community Camp-based Cross-sectional Study" and make related changes wherever possible accordingly.

Response
Thank you. The title and all related changes have been made as per your advice. The words “population-based have been replaced by community camp-based”

Declaration of competing interests:
I declare that I have no competing interests

Response
Thank you.